Author's response to reviews

Title: Can the FAST and ROSIER adult stroke recognition tools be applied to confirmed childhood arterial ischemic stroke?

Authors:

Adriana Yock-Corales (adriyock@gmail.com)
Franz E Babl (Franz.Babl@rch.org.au)
Ian T Mosley (ian.mosley@monash.edu)
Mark T Mackay (Mark.MacKay@rch.org.au)

Version: 3 Date: 17 October 2011

Author's response to reviews: see over
BMC Pediatrics, MS: 1644163769545734

CHANGES AS REQUESTED

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution
DONE

Abstract: this does not have the correct structure, the first section should be a 'Backgrounds' section, not 'Objectives'. Please remember to update the abstract in the submission system.
DONE

Authors' contributions: Please indicate that all authors read and approved the final manuscript. Please format this section as a paragraph, not as a list.
DONE

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].'
DONE

Minor revisions

Author list: please remove author titles and qualifications.
DONE

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efg@generic.co.uk
IJ: IJKL@corporation.com
DONE

Funding: Please move the statement of funding to the Acknowledgements section and remove it from the title page
DONE

Structure: abbreviations should be placed after the Conclusion. Please change 'Introduction' to 'Background'. Please change 'Materials and Methods' to 'Methods'. Please place the Acknowledgements after Authors' Contributions.
DONE

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
DONE
Tables: please remove the visible vertical lines from your tables.
DONE